Author’s response to reviews

Title: Women’s perspectives on caesarean section recovery, infection and the PREPS Trial: A qualitative pilot study

Authors:

Annalise Weckesser (Annalise.weckesser@bcu.ac.uk)
Nicola Farmer (Nicola.Farmer@bwnft.nhs.uk)
Rinita Dam (Rinita.Dam@bcu.ac.uk)
Amie Wilson (A.Wilson.5@bham.ac.uk)
victoria hodgetts morton (v.a.h.morton@bham.ac.uk)
R.Katie Morris (r.k.morris@bham.ac.uk)

Version: 2 Date: 26 Jun 2019

Author’s response to reviews:

Thank you for the comments submitted.